Author’s response to reviews

Title: Multi-parametric effect in predicting tumor histological grade by using Susceptibility Weighted Magnetic Resonance Imaging in tongue squamous cell carcinoma

Authors:

Xing yang (jamieyx@163.com)

jinyu zhu (zjy1012@126.com)

Yongming dai (dymdym118@163.com)

zhen tian (tian0304_cn@163.com)

gongxin yang (yanggongxin1986@163.com)

huimin shi (2240542827@qq.com)

yingwei wu (wuyw0103@hotmail.com)

Xiaofeng Tao (nanxinhao@126.com)

Version: 1 Date: 13 Jul 2018

Author’s response to reviews:

As required, we have revised the figure legend, adjusted the author orders in system to match that in the manuscript and also added the IRB names in the revised manuscript.